Author’s response to reviews

Title: Obese adolescents exhibit a constant ratio of GH isoforms after whole body vibration and maximal voluntary contractions

Authors:

Antonello Rigamonti (antonello.rigamonti@unimi.it)

Michael Haenelt (Michael.Haenelt@med.uni-muenchen.de)

Martin Bidlingmaier (Martin.Bidlingmaier@med.uni-muenchen.de)

Alessandra De Col (a.decol@auxologico.it)

Sofia Tamini (sofia.tamini@gmail.com)

Gabriella Tringali (gabriella.tringali@yahoo.it)

Roberta De Micheli (roby88demicheli@gmail.com)

Laura Abbruzzese (l.abbruzzese@auxologico.it)

Cintia Goncalves da Cruz (ci.renata@gmail.com)

Mario Bernardo-Filho (bernardofilhom@gmail.com)

Silvano Cella (silvano.cella@unimi.it)

Alessandro Sartorio (sartorio@auxologico.it)

Version: 1 Date: 17 Oct 2018

Author’s response to reviews:

We thank the Editor(s) and the Editorial Office for the preliminary evaluation of our manuscript.

1. Overlap

As suggested, we have reworded the Methods due to a presumptive overlap with one of our previous works (ref. [15]). This has been a difficult task, being the exercise protocols particularly detailed and standardised. We would like to ensure that the remaining parts (Background, Results, Discussion and Conclusions) are original.

2. Title page:
We have included the email addresses for all authors on the title page. The corresponding author is clearly indicated.

3. Headings and formatting:

We have better specified the following headings within our manuscript: Background, Methods, Results, Discussion, Conclusions.

4. Authors’ contributions:

We have represented authors' names using their full initials, not their full name, in the Authors’ Contributions section.